Author's response to reviews

Title: Prevalence and factors associated with wound colonization by Staphylococcus spp. and Staphylococcus aureus in hospitalized patients in inland northeastern Brazil: a cross-sectional study

Authors:

Gilmara Celli Maia Almeida (gilmaracelli@yahoo.com.br)
Marquiony Marques Santos (marquiony@gmail.com)
Nara Grazieli Martins Lima (narinhamartins@hotmail.com)
Thiago André Cidral (thiagoandrecidal@gmail.com)
Maria Celeste Nunes Melo (celmelo@gmail.com)
Kenio Costa Lima (limke@uol.com.br)

Version: 4 Date: 2 June 2014

Author's response to reviews:

Dear Editor and Reviewers,

The review is described in a cover letter giving a point-by-point response to the concerns (cover letter attached). Furthermore, attached is the certificate of the English revision.

The changes required are in red text. We tried to answer as best as possible all the recommendations. The changes from the previous revision were placed in normal text color (black). Please do not hesitate to ask for any modification. Sorry in advance if we got it wrong any suggestions, and because of that we left some pending. Hopefully we have reached the aim of leaving the article fit for publication.

Gratefully,

Gilmara Almeida